Author's response to reviews

Title: The perception of healthcare quality of the elderly in the city of Bari, South Italy.

Authors:

Rosa Prato (r.prato@unifg.it)
Domenico Martinelli (domenicomartinelli@gmail.com)
Annarita Fusco (specializzandi@igiene.uniba.it)
Annarita Panebianco (specializzandi@igiene.uniba.it)
Pietro L Lopalco (Perluigi.Lopalco@ecdc.eu.int)
cinzia A Germinario (c.germinario@igiene.uniba.it)
Michele Quarto (m.quarto@igiene.uniba.it)

Version: 3 Date: 17 October 2007

Author's response to reviews:

Prof. C.A. Germinario
Sezione di Igiene  DIMO
University of Bari
Policlinico, Piazza G. Cesare
I-70124 Bari, Italy
Tel: +39 (0)80 5478484
Fax: +39 (0)80 5478472
Email: c.germinario@igiene.uniba.it

BMC Health Services Research (Editor)
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK

Bari, October, 17th 2007

I submit to your attention the following changes to the format of manuscript:
- Title  The full stop has been removed from the end of the title.
- Affiliations  The affiliations have been written in full.
- 'Abstract' has been included at the top of page 2.
- We have uploaded the new abstract version in the submission system.
- Footnotes  we incorporated these into the text of the manuscript.
- Competing interests: The authors have no competitive interests. So we declared in the section.
- Authors contributions has been changed in Authors' contributions.
- The term "home care services integrated" has been implemented in the text, like suggested by reviewer Marten Lagergren.

With my best regards.

Cinzia Germinario